Author's response to reviews

Title: Positive Association between Aspirin-Intolerant Asthma and Genetic Polymorphisms of FSIP1: a Case-Case Study

Authors:

Jason Yongha Kim (imblue870227@gmail.com)
Jeong-Hyun Kim (soykm@sogang.ac.kr)
Tae Joon Park (pires81@naver.com)
Joon Seol Bae (jsbae@sogang.ac.kr)
Jin Sol Lee (shadow2386@naver.com)
Charisse Flerida A Pasaje (charissepass@yahoo.com)
Byung Lae Park (blpark@snp-genetics.com)
Hyun Sub Cheong (chhs@snp-genetics.com)
Jong-Sook Park (js1221@schbc.ac.kr)
Sung-Woo Park (swpark@schbc.ac.kr)
Soo-Taek Uh (uhs@hosp.sch.ac.kr)
Mi-Kyeong Kim (kimmk@chungbuk.ac.kr)
Inseon S Choi (ischoi@chonnam.ac.kr)
Sang Heon Cho (shcho@plaza.snu.ac.kr)
Byoung Whui Choi (bwchoimd@cau.ac.kr)
Choon-Sik Park (mdcspark@unitel.co.kr)
Hyoung-Doo Shin (hdshin@sogang.ac.kr)

Version: 5 Date: 25 May 2010

Author's response to reviews: see over
Major revisions (we require the author to make these changes)

Additional files: Please ensure that additional files are referred to as 'Additional file x' throughout the manuscript and cited in numerical order.

→ Reply: We have changed reference to additional files as you requested. Subsequently, we changed the order of additional files submitted as they appear on the manuscript.

Links: Please remove this section. URLs should either be cited in the reference list or the main manuscript text.

→ Reply: We have removed the links section. SNPSpD link is provided in reference section and hapmap link is provided in the main manuscript text.